Author's response to reviews

Title: Healthy eating and obesity prevention for preschoolers: a randomised controlled trial

Authors:

Helen Skouteris (helens@deakin.edu.au)  
Marita McCabe (marita.mccabe@deakin.edu.au)  
Boyd Swinburn (boyd.swinburn@deakin.edu.au)  
Briony Hill (briony.hill@deakin.edu.au)

Version: 2  Date: 29 March 2010

Author's response to reviews: see over
March 29, 2010

Melissa Norton, MD
Editor-in-Chief, Biomed Central Public Health

Dear Doctor Melissa Norton,

Subject: MS: 1511856226365957

Thank you for provisionally accepting our manuscript entitled “Healthy eating and obesity prevention for preschoolers: a randomised controlled trial”.

I would like to advise of the following changes:

(1) The statement “This study was approved by the Deakin University Human Research Ethics Committee on the 7th December 2009 (DU-HREC 2009-180)” has been included at the end of the first paragraph of the ‘Design and Methods’ section of the manuscript.

I look forward to hearing back from you in due course.

Yours sincerely,

Helen

Dr Helen Skouteris
School of Psychology,
Deakin University, Victoria, 3125.
Email: helens@deakin.edu.au